Author’s response to reviews

Title: Elevated levels of plasma D-dimer predict a worse outcome in patients with nasopharyngeal carcinoma

Authors:

WenHui Chen (chenwh@sysucc.org.cn)
LinQuan Tang (tanglq@sysucc.org.cn)
FengWei Wang (wangfengw@sysucc.org.cn)
ChangPeng Li (lizhp@sysucc.org.cn)
XiaoPeng Tian (tianxp@sysucc.org.cn)
XiaoXia Huang (huangxiaox@sysucc.org.cn)
ShiJuan Mai (maishj@sysucc.org.cn)
YiJi Liao (liaoyj@sysucc.org.cn)
HaiXia Deng (denghx@sysucc.org.cn)
QiuYan Chen (chenyq@sysucc.org.cn)
Huai Liu (liuhuai@sysucc.org.cn)
Lu Zhang (zhanglu@sysucc.org.cn)
ShanShan Guo (guoshsh@sysucc.org.cn)
LiTing Liu (ljult@sysucc.org.cn)
ShuMei Yan (yanshm@sysucc.org.cn)
ChaoFeng Li (lichaofeng@sysucc.org.cn)
JingPing Zhang (zhangjp@sysucc.org.cn)
Qing Liu (liuqing@sysucc.org.cn)
XueWen Liu (liuxw@sysucc.org.cn)
LiZhi Liu (liulizh@sysucc.org.cn)
HaiQiang Mai (maihq@sysucc.org.cn)
MuSheng Zeng (zengmsh@sysucc.org.cn)
Dan Xie (xied@mail.sysu.edu.cn)

Version: 4 Date: 30 June 2014

Author’s response to reviews: see over
Dear Dr. Bravo,

We are pleased to submit a revised manuscript again, entitled “Elevated levels of plasma D-dimer predict a worse outcome in patients with nasopharyngeal carcinoma” (Revision MS: 1465528363128895). We greatly appreciate that your constructive suggestion. All authors have read and approved the revised manuscript, and we have revised manuscript conforms to the journal style of BMC cancer. There are no financial or other relationships that might lead to a conflict of interest.

I hope that the revised manuscript will be accepted for publication. Below, we have restated the comments of editor and provided point-by-point responses. The changes are highlighted in red in the revised manuscript.

Annie Lyn Bravo
June 30, 2014
BMC cancer
Editor's Comment:

"Two minor comments:

1. An interesting paper can be referred to as below:


Reply: Thank you for your constructive suggestions. This is an interesting paper for plasma EBV DNA, we have referred as reference 38.

Revised in reference 38.

2. The below one is more representing, which using evidence-based approach, can replace the reference #37.


Reply: We appreciate the editor’s suggestion. We are totally agree with that this paper is a more representing and evidence-based. We have replaced reference 37 with the paper mentioned above.

Revised in reference 37.

We look forward to your favourable reply.

Sincerely yours,
Dan Xie, M.D., Ph.D.

Head, the Department of Pathology,

Principal Investigator, the State Key Laboratory of Oncology in Southern China,

Sun Yat-Sen University Cancer Centre,

No. 651, Dongfeng Road East, 510060 Guangzhou, China

Tel: 86-20-87343193; Fax: 86-20-87343170

E-mail: xied@mail.sysu.edu.cn